Author’s response to reviews

Title: Single-stage transforaminal decompression, debridement, interbody fusion, and posterior instrumentation for lumbosacral brucellosis

Authors:
yakefu abulizi (yakup0213@126.com)
weidong liang (362746037@qq.com)
aikeremujing muheremu (yakupspine@163.com)
weibin sheng (wbsheng@vip.sina.com)
Maierdan Maimaiti (halida0222@163.com)

Version: 4 Date: 11 Jun 2017

Author’s response to reviews:
Thank you for your letter for our manuscript entitled “Single-stage transforaminal decompression, debridement, interbody fusion, and posterior instrumentation for lumbosacral brucellosis” (BSUR-D-17-00077).

1. We have corrected the "Authors’ contribution" section according to the guidelines for authorship.

2. All the figures, except Fig 2a-b, have referenced in the main text. But we didn`t remove these two figures because they are related to the Table 2. Anyway, we agree with any change of our figures by your production department.

We appreciate for editors` warm work earnestly, and hope that the correction will meet with approval.